Author's response to reviews

Title: Acceptability of Medical Male Circumcision in the Traditionally Circumcising Communities in Northern Tanzania

Authors: 

Mwita Wambura (wmwita@yahoo.com)
Joseph R Mwanga (jrmwanga@yahoo.co.uk)
Jacklin F Mosha (jfmosha@yahoo.com)
Gerry Mshana (gmshana@nimr.or.tz)
Frank Mosha (fmosha2002@yahoo.co.uk)
John Changalucha (jchangalucha@yahoo.com)

Version: 5 Date: 3 May 2011

Author's response to reviews:

National Institute for Medical Research,

PO Box 1462,

Mwanza, Tanzania.

17th April 2011

Editor,

BMC Public Health,

BioMed Central Ltd,

Floor 6,

236 Gray’s Inn Road,

London, WC1X 8HB,

United Kingdom

Re: ACCEPTABILITY OF MEDICAL MALE CIRCUMCISION IN THE TRADITIONALLY CIRCUMCISING COMMUNITIES IN NORTHERN TANZANIA
We are pleased to re-submit the attached manuscript for consideration by the BMC public health. This paper has been reviewed following reviewers’ comments. Attached with this cover letter, please, find the revised manuscript and our responses to reviewers’ comments.

We have read the editorial policies. These data have not been submitted for publication elsewhere and is not being considered for publication elsewhere. There are no conflicts of interest for this paper. All appropriate ethics approval has been obtained for the research reported. All authors have contributed to and approved this paper.

We look forward to hearing from you.

Sincerely,

Mwita Wambura
Joseph Mwanga
Jackline Mosha
Gerry Mshana
Frank Mosha
John changalucha